Author’s response to reviews

Title: Script Concordance Tests: Guidelines for Construction

Authors:

Jean Paul Fournier (Fournier.jp@chu-nice.fr)
Anne Demeester (Anne.Demeester@mail.ap-hm.fr)
Bernard Charlin (bernard.charlin@umontreal.ca)

Version: 2 Date: 28 November 2007

Author’s response to reviews:

It is the manuscript submitted to BMC Biology. No modification has been made.